Author's response to reviews

Title: Radiation dose reduction at a price: the effectiveness of a male gonadal shield during helical CT scans

Authors:

Lawrence T Dauer (dauerl@mskcc.org)
Kevin A Casciotta (casciotk@mskcc.org)
Yusuf E Erdi (erdiy@mskcc.org)
Lawrence N Rothenberg (rothenbl@mskcc.org)

Version: 4 Date: 15 March 2007

Author's response to reviews:

This revision addresses the changes requested by Senior Assistant Editor, Iratxe Puebla, in email dated 3-14-07. Thank you for your assistance. Lawrence T. Dauer, PhD, CHP.